Author's response to reviews

Title: Promoting male involvement to improve PMTCT uptake and reduce antenatal HIV infection: a cluster randomized control trial protocol

Authors:

Karl Peltzer (kpeltzer@hsrc.ac.za)
Deborah Jones (DJones@med.miami.edu)
Stephen Weiss (sweiss2@med.miami.edu)
Elisa Shikwane (eshikwane@hsrc.ac.za)

Version: 2 Date: 28 September 2011

Author's response to reviews:

(1) Please remove the reference citations from the abstract of your manuscript (please ensure this is also done in the abstract within the submission system)

Done as below

Background: Despite the availability of a dual therapy treatment protocol and infant feeding guidelines designed to prevent mother to child transmission (PMTCT) of HIV, of the over 1 million babies born in South Africa each year, only 70% of those born to HIV positive mothers receive dual therapy. Similar to other resource-poor nations facing the integration of PMTCT into routine pregnancy and infant care, efforts in South Africa to scale up PMTCT and reduce transmission to <5% have fallen far short of the United Nation’s goal of 50% reductions in paediatric HIV by 80% coverage of mothers.

(2) Include the email addresses of all authors within the title page of your manuscript.

Email addresses:
KP: kpeltzer@hsrc.ac.za
DJ: djones@med.miami.edu
SW: sweiss2@med.miami.edu
ES: eshikwane@hsrc.ac.za

(2) Include a ‘Competing interests’ section between the Conclusions and Authors’ contributions. If there are none to declare, please write ‘The authors declare that they have no competing interests’.

Competing interests
The authors declare that they have no competing interests.
Authors' contributions

KP, DJ and SW were the main contributors to the conceptualization of the study. KP and DJ also contributed significantly to the first draft of the paper and all authors contributed to the subsequent drafts and finalization. All authors read and approved the final manuscript.